MATHEMATICS 300 — FALL 1997 Introduction to Mathematical Reasoning

Instructor: H. J. Sussmann HANDOUT NO. 6. November 10, 1997

INSTRUCTOR’S OFFICE HOURS

Monday and Thursday, 4th period. Please let the instructor know after class if you are planning to go see him during office hours, so he will not have to stay in his office unless he knows somebody is coming.

TEACHING ASSISTANT’S OFFICE HOURS

Friday 11:30 a.m.- 2:00 p.m.. Hill 101. (The TA’s name is David Radnell.)

HOMEWORK ASSIGNMENT NO. 6 Due on Thursday, November 13:

Problems 2a,c,d,e,f 3, 4, 13, 16b,c, 18a,c c, 19, in Pages 111-114.

READING FOR NOVEMBER 10-13 1. Finish Chapter 2, especially Sections 2.3 and 2.6 2. Chapter 3, Section 1.

3. Read carefully the transcript “NOVA #2414: The Proof.”

The SECOND MIDTERM EXAM will be on MONDAY, DECEMBER 1.

The FINAL EXAM will be on FRIDAY, DECEMBER 19, in TCB-107, from 8:00 a.m. to 11:00 a.m.